Author's response to reviews

Title: The botanical integrity of wheat products influences the gastric distention and satiety in healthy subjects

Authors:

Joanna Hlebowicz (Joanna.Hlebowicz@med.lu.se)
Sandra Lindstedt (Sandra.Lindstedt@skane.se)
Peter Hoglund (Peter.Hoglund@skane.se)
Ola Bjorgell (Ola.Bjorgell@med.lu.se)
Lars-Olof Almer (Lars-Olof.Almer@med.lu.se)
Gassan Darwiche (Gassan.Darwiche@skane.se)

Version: 6 Date: 13 April 2008

Author's response to reviews:

Dear Nutrition Journal Editorial Team Malmö 2008-04-13

Please notice that the changes have been done as required:
• The trial registration has been added to the abstract.
• The highlighting/tracking has been removed from the manuscript.
• Page 9, the website address has been removed from the text.
• Figure legends are repositioned after the references and the symbols are changed into descriptive text instead.
• Figure 1 and 3 are cropped to minimise white space.

Do not hesitate to contact me further if you have any additional questions. I am hoping for further cooperation in the future.

Sincerely,
Joanna Hlebowicz, MD
Dept. of Medicine, Malmö University Hospital
Ingång 35, 205 02 Malmö, Sweden
Telephone nr +46-40-331000
Fax nr +46-40-336208
E-mail: Joanna.Hlebowicz@med.lu.se